\documentclass[a4paper]{panl}
\usepackage{cite}
\usepackage{wrapfig}
\usepackage{graphicx}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{amsmath}
\usepackage{longtable}
\usepackage{rotating}
\usepackage{lscape}
\usepackage{epsfig}
\usepackage{multirow}


\originalTeX
%\russianTeX
\begin{document}
\begin{thebibliography}{1}
\def\selectlanguageifdefined#1{
\expandafter\ifx\csname date#1\endcsname\relax
\else\selectlanguage{#1}\fi}
\providecommand*{\href}[2]{{\small #2}}
\providecommand*{\url}[1]{{\small #1}}
\providecommand*{\BibUrl}[1]{\url{#1}}
\providecommand{\BibAnnote}[1]{}
\providecommand*{\BibEmph}[1]{\emph{#1}}
\ProvideTextCommandDefault{\cyrdash}{\hbox to.8em{--\hss--}}
\providecommand*{\BibDash}{\ifdim\lastskip>0pt\unskip\nobreak\hskip.2em\fi
\cyrdash\hskip.2em\ignorespaces}

\bibitem{Zakharov:2005bc}
\selectlanguageifdefined{english}
\BibEmph{Zakharov, A.F.} {BLACK HOLES: OBSERVATIONAL PROPERTIES}~//
\href{http://dx.doi.org/10.1142/S0217751X05024560}{International Journal of Modern Physics}
 % \BibDash
\newblock 2005. \BibDash
\newblock V.~20. 

\bibitem{Zakharov:2014lqa}
\selectlanguageifdefined{english}
\BibEmph{Zakharov, A.F.} {BLACK HOLES: OBSERVATIONAL PROPERTIES}~//
\href{https://doi.org/10.1103/PhysRevD.90.062007}{Phys. Rev. D}
 % \BibDash
\newblock 2014. \BibDash
\newblock V.~90 \BibDash
\newblock P.~062007.

\bibitem{EventHorizonTelescope:2022apq}
\selectlanguageifdefined{english}
\BibEmph{The Event Horizon Telescope Collaboration} {First Sagittarius A* Event Horizon Telescope Results. VI. Testing the Black Hole Metric}~//
\href{https://ui.adsabs.harvard.edu/link_gateway/2022ApJ...930L..17E/doi:10.3847/2041-8213/ac6756}{Astrophys.J.Lett.}
 % \BibDash
\newblock 2022. \BibDash
\newblock V.~930, no.~2. \BibDash
\newblock P.~L17.

\bibitem{EventHorizonTelescope:2019pgp}
\selectlanguageifdefined{english}
\BibEmph{The Event Horizon Telescope Collaboration} {First M87 Event Horizon Telescope Results. V. Physical Origin of the Asymmetric Ring}~//
\href{https://ui.adsabs.harvard.edu/link_gateway/2019ApJ...875L...5E/doi:10.3847/2041-8213/ab0f43}{Astrophys.J.Lett.}
 % \BibDash
\newblock 2022. \BibDash
\newblock V.~875, no.~1. \BibDash
\newblock P.~L5.

\bibitem{Cui:2023uyb}
\selectlanguageifdefined{english}
\BibEmph{Cui, Y.; others.} {Precessing jet nozzle connecting to a spinning black hole in M87}~//
\href{https://www.nature.com/articles/s41586-023-06479-6}{Nature}
 % \BibDash
\newblock 2023. \BibDash
\newblock V.~621 \BibDash
\newblock P.~711-715.

\bibitem{Newman:1965tw}
\selectlanguageifdefined{english}
\BibEmph{Newman, E.T.; Janis, A.I.} {Note on the Kerr Spinning-Particle Metric}~//
\href{https://doi.org/10.1063/1.1704350}{J. Math. Phys.}
 % \BibDash
\newblock 1965. \BibDash
\newblock V.~6 \BibDash
\newblock P.~915–917.

\bibitem{Azreg-Ainou:2014aqa}
\selectlanguageifdefined{english}
\BibEmph{Azreg-Aïnou, M.} {From static to rotating to conformal static solutions: Rotating imperfect fluid wormholes with(out) electric or magnetic field}~//
\href{https://doi.org/10.1140/epjc/s10052-014-2865-8}{Eur. Phys. J. C}
 % \BibDash
\newblock 2014. \BibDash
\newblock V.~74 \BibDash
\newblock P.~2865.

\bibitem{Capozziello:2023tbo}
\selectlanguageifdefined{english}
\BibEmph{Capozziello, S.; Zare, S.; Hassanabadi, H.} {Testing bumblebee gravity with global monopoles in a dark matter spike by EHT observations from M87 and Sgr A}~//
\href{https://arxiv.org/abs/2311.12896}
 % \BibDash
\newblock arXiv:2311.12896~[hep-th].

\bibitem{Pepan24}
\selectlanguageifdefined{russian}
\BibEmph{Алексеев, С. О., Байдерин, А. А. и Зенин, О. И.} {Тени черных дыр в моделях Хорндески и бамбелби: учет вращения}~//
\href{https://istina.msu.ru/journals/97213/}{ЭЧАЯ}
 % \BibDash
\newblock 2025. \BibDash
\newblock V.~56 no.~2. \BibDash
\newblock P.~0367-2026.

\bibitem{Jetp24}
\selectlanguageifdefined{russian}
\BibEmph{Алексеев С.О., Байдерин А.А., Немтинова А.В., Зенин О.И.} {Нелокальные гравитационные теории и изображения теней черных дыр}~//
\href{http://www.jetp.ras.ru/cgi-bin/r/index/r/165/4/p508?a=list}{ЖЭТФ}
 % \BibDash
\newblock 2024. \BibDash
\newblock V.~165 no.~4. \BibDash
\newblock P.~508.

\bibitem{dambrosio2021black}
\selectlanguageifdefined{english}
\BibEmph{Fabio D'Ambrosio, Shaun D.B. Fell, Lavinia Heisenberg, Simon Kuhn} {Black holes in gravity}~//
\href{https://doi.org/10.1103/PhysRevD.105.024042}{Phys. Rev. D}
 % \BibDash
\newblock 2022. \BibDash
\newblock V.~105 no.~2. \BibDash
\newblock P.~024042.

\bibitem{Prokopov:2021lat}
\selectlanguageifdefined{english}
\BibEmph{Prokopov, V.A.; Alexeyev, S.O.; Zenin, O.I.} {Black Hole Shadows Constrain Extended Gravity}~//
\href{https://doi.org/10.1134/S1063776122070093}{J. Exp. Theor. Phys.}
 % \BibDash
\newblock 2022. \BibDash
\newblock V.~135 \BibDash
\newblock P.~91–99.

\bibitem{Zenin2}
\selectlanguageifdefined{english}
\BibEmph{Prokopov, V.A.; Alexeyev, S.O.; Zenin, O.I.} {Black Hole Shadows Constrain Extended Gravity 2: Sgr A*}~//
\href{http://dx.doi.org/10.1134/S1063776122120172}{J. Exp. Theor. Phys.}
 % \BibDash
\newblock 2022. \BibDash
\newblock V.~135 \BibDash
\newblock P.~842–843.

\bibitem{De_Lorenzo_2015}
\selectlanguageifdefined{english}
\BibEmph{Tommaso De Lorenzo, Costantino Pacilio, Carlo Rovelli, Simone Speziale} {On the Effective Metric of a Planck Star}~//
\href{https://doi.org/10.1007/s10714-015-1882-8}{Gen. Rel. Grav}
 % \BibDash
\newblock 2015. \BibDash
\newblock V.~47 no.~4. \BibDash
\newblock P.~1882.


\end{thebibliography}

% Journal sections (see http://pkp.jinr.ru/index.php/PEPAN_LETTERS/about/editorialPolicies#focusAndScope)
%\issuearea{Physics of Elementary Particles and Atomic Nuclei. Theory}
% or in Russian
%\issuearea{ФИЗИКА ЭЛЕМЕНТАРНЫХ ЧАСТИЦ И АТОМНОГО ЯДРА. ТЕОРИЯ}

\title{Particular notes in Black Hole shadow's modelling when spinning is taken into account \\ Особенности моделирования теней черных дыр при учете вращения}
\maketitle
\authors{O.\,Zenin$^{a,}$\footnote{E-mail: dkiiiabu4@gmail.com},
S.\,Alexeyev$^{a,b,}$\footnote{E-mail: alexeyev@physics.msu.ru},
A.\,Baiderin$^{a,}$\footnote{E-mail: baiderin21a@gmail.com}}
\setcounter{footnote}{0}
\authors{О.И.\,Зенин$^{a}$,
С.О.\,Алексеев$^{a,b}$,
А.А.\,Байдерин$^{a,b}$}
\from{$^{a}$\,Department of Quantum Theory and High Energy Physics, Physics Faculty, Lomonosov Moscow State University,\\ Vorobievi Gory, 1/2, Moscow, 117234, Russia}
\from{$^{a}$\,Кафедра квантовой теории и физики высоких энергий, Физический факультет, Московский государственный университет имени М.В. Ломоносова, Воробьёвы Горы, 1/2, Москва, 119234, Россия}
\from{$^{b}$\,Sternberg Astronomical Institute, Lomonosov Moscow State University,\\ Univeresitesky Prospekt, 13, Moscow, 117234, Russia}
\from{$^{b}$\,Государственный астрономический институт имени П.К. Штернберга, Московский государственный университет имени М.В. Ломоносова, Университетский проспект, 13, Москва, 119234, Россия}

\begin{abstract}
% Russian translation of the abstract

Обсуждаются особенности улучшенного алгоритма Ньюмена-Яниса и его применение для получения метрик черных дыр в расширенных теориях гравитации. Рассмотрены теории  расширяют ОТО различными способами: петлевая квантовая гравитация и $f(Q)$ гравитация (симметричная телепараллельная гравитация STEGR).\
\vspace{0.2cm}

We discuss specific aspects of the improved Newman-Janis algorithm and its application to generate rotating black hole metrics in the extended theories of gravity. We consider here different theories extending general relativity in various ways: loop quantum gravity and $f(Q)$ gravity (symmetric teleparallel gravity STEGR).
\end{abstract}
\vspace*{6pt}

\noindent
PACS: 44.25.$+$f; 44.90.$+$c

\label{sec:intro}
\section*{Introduction}

Since the pioneering works on black hole shadows \cite{Zakharov:2005bc} it has been realized that accounting of the tidal charge in the Reissner-Nordström metric may allow to measure the contribution of new physics extending the general relativity (GR) \cite{Zakharov:2014lqa}. To proceed, one must apply the new observational black hole images. So, comparing the results of shadow modeling in various gravity models one could find the model describing the shadow, accretion disk, ... better than the others. At the first stage only the spherically-symmetric solutions were used. However, both black holes photographed by the Event Horizon Telescope (EHT) (M87 and Sgr A) rotate \cite{EventHorizonTelescope:2022apq,EventHorizonTelescope:2019pgp} (the angular velocity was measured quite recently \cite{Cui:2023uyb}). Therefore, to increase the accuracy it is highly desirable to take into account the effects of rotation. That is why one has to make a transition from Reissner-Nordstrom-type metrics to Kerr-Newman-type ones. In general the direct solving Einstein's equations for axially symmetric spacetime leads to complex implicit partial differential equations. Thus, it is preferable to find alternative methods. For example one can generate new solutions by introducing rotation in the parametric space. Therefore, making such a turn from the case $a=0$, where $a$ is the rotational parameter of a black hole, to $a\neq 0$, the Newman-Janis algorithm \cite{Newman:1965tw} allows for obtaining rotating solutions in a simpler way. The algorithm was later refined to facilitate this transformation in a simpler way \cite{Azreg-Ainou:2014aqa}. This is important for studying BH shadows in extended theories of gravity (for example, Horndeski, bumblebee) \cite{Capozziello:2023tbo}, where rotation significantly influences the shape of the shadow.

\label{sec:Rotating solutions}
\section*{Rotating solutions}

Previously, the Newman-Janis algorithm \cite{Newman:1965tw} in its improved version \cite{Azreg-Ainou:2014aqa} has been used in \cite{Pepan24,Jetp24}. Thus, the process begins by considering the spherically symmetric metric in the form:
\begin{equation}\label{metric} 
ds^2 =  - G(r) dt^2 + \frac{1}{F(r)} dr^2 + H(r)d\Omega^2,
\end{equation}
where $G(r)$, $F(r)$ and $H(r)$ are the metric functions. Such a representation makes possible the application of the wide class of spherically symmetric space-times. The components of the axially symmetric metric obtained from equation \eqref{metric} in \cite{Azreg-Ainou:2014aqa} are:
\begin{eqnarray}\label{spin_metri}
g_{tt} & = & -\frac{FH + a^2\cos^2{\theta}}{(K + a^2\cos^2{\theta})^2} \Psi, \qquad g_{t\phi}  =  -a\sin^2{\theta}\frac{K - FH}{(K + a^2\cos^2{\theta})^2}\Psi,\nonumber \\
g_{\theta \theta} & = & \Psi,\qquad g_{rr}  =  \frac{\Psi}{FH + a^2}, \\
g_{\phi \phi} & = & \Psi\sin^2{\theta}(1 + a^2\sin^2{\theta}\frac{2K - FH + a^2\cos^2{\theta}}{(K + a^2\cos^2{\theta})^2}), \nonumber
\end{eqnarray}
where $K = H(r)\sqrt{F(r)/G(r)}$.

In the next step, one introduce the function $\Psi(r,y^2,a)$($y\equiv\cos{\theta}$) satisfying the conditions:
\begin{eqnarray}\label{dif_eq}
&& \lim_{a\to 0} \Psi(r,y^2,a) = H(r), \\
&& (K + a^2y^2)^2(3\Psi_r\Psi_{y^2} -2\Psi\Psi_{r,y^2}) = 3a^2K_r\Psi^2,\nonumber\\
&& \Psi \left[ K^2_r + K(2 - K_{rr}) -a^2y^2(2 + K_{rr})\right] + (K + a^2y^2)[(4y^2\Psi_{y^2} - K_r\Psi_r] = 0. \nonumber
\end{eqnarray}
The first line in eq. (\ref{dif_eq}) describes the behavior at $a\to 0$, being the reverse transition to the non-rotating metric with two representations ($\Psi_n$ and $\Psi_c$) connected by the conformal transformation. So the initial metric is: $ds_{c}^{2} = \Psi_c/\Psi_nds^2_n$ and thus, one finds the solution of eq. (\ref{dif_eq}) in the form:
\begin{eqnarray}
&& \Psi_c  =  H(r)\exp{[a^2f(r,a^2y^2,a)]} \approx  H(r) + a^2X(y^2,r) + o(a^2), \\
&& X(y^2,r)  =  \frac{H^2(8K - K_{r}^2)y^2}{K^2(8H - H_rK_r)}. \nonumber
\end{eqnarray}

\label{sec:f(Q) gravity}
\section*{f(Q) gravity}

We consider the Symmetric Teleparallel Equivalent of General Relativity (STEGR) with a non-zero non-metricity scalar $Q$ \cite{dambrosio2021black}. The metric extending GR ($I^{+}$) has the form \cite{dambrosio2021black,Prokopov:2021lat}:
\begin{eqnarray}\label{f_Q}
G(r) = 1 - \frac{2M_{ren}}{r}  + \alpha\frac{32}{r^2}, \qquad
F(r) = 1 - \frac{2M_{ren}}{r}  + \alpha\frac{96}{r^2}, \qquad
H(r) = r^2,
\end{eqnarray}
where $\alpha$ is the expansion parameter, $c_1$ is the integration constant, $M_{ren}$ is the re-normalized mass. Note that a far observer cannot detect the difference between re-normalized and ordinary Schwarzschild masses. So we use $M_{ren}=M$ everywhere, normalizing all values to it. In \cite{Zenin2} the restrictions on $\alpha$ ($-0.008<\alpha<0.005$) were obtained at $M=1$. Therefore, we assume that $\left|\alpha \right| \ll M$. Also note that in our case $r>2M$ (event horizon in GR). We begin the calculations with $\sqrt{F/G}$:
\begin{eqnarray}
\sqrt{\frac{F}{G}}=\frac{1 - \frac{2M_{ren}}{r}  + \alpha\frac{32}{r^2}+\alpha\frac{64}{r^2}}{1 - \frac{2M_{ren}}{r}  + \alpha\frac{32}{r^2}} = 1+\frac{64\alpha}{r^2-2Mr+32\alpha} \approx 1.
\end{eqnarray}
So $K=r^2$ and $K_r=2r$. The metric in such approximation is symmetric if $F=G$ with the last term $96\alpha/r^2$. In this case $X(y^2,r)=y^2=\cos^2{\theta}$ and $\Psi=r^2+a^2\cos^2{\theta}$. As a result, we denote $A^2=-2Mr+96\alpha$, $\rho^2=r^2 + a^2\cos^2\theta$ obtaining:
\begin{eqnarray}\label{spin_metri_fQ}
g_{tt}  =  -\frac{\rho^2 + A^2}{\rho^2}, \qquad g_{t\phi}  =  a\sin^2{\theta}\frac{A^2}{\rho^2}, \qquad g_{\theta \theta}  =  \rho^2, \nonumber\\
g_{rr}  =  \frac{\rho^2}{r^2 + a^2  + A^2}, \qquad g_{\phi \phi}  =  \sin^2{\theta}\left( \rho^2 + a^2\sin^2{\theta}\frac{\rho^2 - A^2}{\rho^2}\right).
\end{eqnarray}

\label{sec:Loop Quantum Gravity}
\section*{Loop Quantum Gravity}

We consider the modified Hayward metric without central singularity \cite{De_Lorenzo_2015,Prokopov:2021lat} where:
\begin{eqnarray}\label{loop} 
G(r) & = & \left(1 - \frac{2Mr^2}{r^3+2Ml^2}\right)\left(1 - \frac{M\alpha\beta}{\alpha r^3+\beta M}\right), \nonumber \\
F(r) & = & 1 - \frac{2Mr^2}{r^3+2Ml^2}, \qquad H(r)  =  r^2, 
\end{eqnarray}
where the parameter $l$ measures the central energy density with $3/8\pi^2$, $\alpha$ is the time delay between the center and infinity, and $\beta$ is associated with single-loop quantum corrections to the Newtonian potential. In \cite{De_Lorenzo_2015} the parameters were constrained as follows: $0\leq\alpha<1$, $\beta_{max}=41/(10\pi)\approx1.305$. For $l > \sqrt{16/27} M \approx 0.7698M$ the object has no horizon. In \cite{Prokopov:2021lat} it was shown that these values fall under restrictions from EHT data. Therefore, the additional parameters in this model cannot be small relative to $M$. Next:
\begin{eqnarray}
\sqrt{\frac{F}{G}}=\Biggl( 1-\frac{M\alpha\beta}{\alpha r^3+\beta M} \Biggr)^{-1/2} = A \neq 1.
\end{eqnarray}
From this expression one concludes that it cannot be equal to $1$. Therefore:
\begin{eqnarray}
K=r^2A \neq r^2, \qquad K_r= A \Biggl( 2r-\frac{3r^4 M\alpha^2 \beta}{2(\alpha^3+\beta M} \Biggr).
\end{eqnarray}.

Further $X$ ($\rho^2=r^2 + a^2\cos^2\theta$): $X=y^2/A=\cos^2\theta/A$. $\Psi=r^2+a^2\cos^2{\theta}/A$. Let $B=r^2A+a^2\cos^2{\theta}$ and $C=2Mr^4(r^3+2Ml^2)^{-1}$ so the new components are:
\begin{eqnarray}\label{spin_metri_lqg}
&& g_{tt}  =  -\frac{\rho^2 -C}{AB}, \qquad 
g_{t\phi}  =  a\sin^2{\theta}\frac{r^2(A-1)+C}{AB}, \nonumber \\ 
&& g_{\theta \theta}  =  \frac{B}{A}, \qquad 
g_{rr}  =  \frac{B}{A(r^2+a^2-C)}, \nonumber \\
&& g_{\phi \phi}  =  \frac{B}{A}\sin^2{\theta}\left( 1 + a^2\sin^2{\theta}\frac{r^2(2A-1)+C+a^2\cos^2{\theta}}{B^2}\right). 
\end{eqnarray}
In opposite when $a=0$, the following static metric is realized:
\begin{eqnarray}\label{spin_metri_lqg}
g_{tt}  =  -G, \qquad g_{t\phi}  =  0, \qquad g_{\theta \theta}  =  r^2, \qquad 
g_{rr}  =  \frac{1}{F}, \qquad g_{\phi \phi}  =  r^2\sin^2{\theta}.
\end{eqnarray}
Therefore, here even at $\Psi \neq r^2+a^2\cos^2{\theta}$ the exact transition leads to the original static metric at $a=0$.

\label{sec:Conclusion}
\section*{Conclusion}

We discuss the specifics of the improved Newman-Janis algorithm and show how to apply it to obtain rotating solutions for $f(Q)$ gravity and loop quantum gravity. For $f(Q)$ gravity the rotating metric transits to the symmetric one at $a=0$ (although initially the metric was not such). In contrast, for loop quantum gravity the rotating metric at $a=0$ turns into the original asymmetric one.

\label{sec:funding}
\section*{Funding}
The work of Oleg Zenin was supported in part by the Foundation for the Advancement of Theoretical Physics and Mathematics ``BASIS'' grant number 22-2-2-11-1. The study was conducted under the state assignment of Lomonosov Moscow State University.


%\nocite{*}
%\bibliographystyle{pepan}
%\bibliography{maikbibl}
% Bibliography automatically generated via BibTeX (see template_bibtex.tex)




\end{document}